Author's response to reviews

Title: Evaluating the SOS Suicide Prevention Program: A Replication and Extension

Authors:

Robert H Aseltine Jr (aseltine@uchc.edu)
Amy James (ajames@uchc.edu)
Elizabeth A Schilling (eschilling@uchc.edu)
Jaime Glanovsky (jglanovsky@uchc.edu)

Version: 2 Date: 23 October 2006

Author's response to reviews:

Updating clinical trials registration information.